Author’s response to reviews

Title: Effect of paracetamol (acetaminophen) and ibuprofen on body temperature in acute ischemic stroke. PISA, a phase II double blind randomized placebo-controlled trial. ISRCTN 98608690

Authors:

Dr Diederik WJ Dippel (dippel@neuro.fgg.eur.nl)
Eric J van Breda (breda@neuro.fgg.eur.nl)
Dr H Maarten A van Gemert (vgemert@worldonline.nl)
H Bart van der Worp (h.b.vandeworp@neuro.azu.nl)
Ron J Meijer (meijer@neur.azr.nl)
L Jaap Kappelle (l.kappelle@neuro.azu.nl)
Peter J Koudstaal (koudstaal@neur.azr.nl)

Version: 2 Date: 15 Jan 2003


Dear editors,

Hereby we resubmit our paper: "Effect of paracetamol (acetaminophen) and ibuprofen on body temperature in acute ischemic stroke. PISA, a phase II double blind randomized placebo-controlled trial (ISRCTN 98608690)" to BMC Cardiovascular Disorders.

We are confident that we have been able to deal with comments of the two reviewers. Their comments were very helpful.

All changes in the manuscript have been marked yellow.

Yours sincerely,

Diederik Dippe